# 4 Columbia Business School

# CENTER ON JAPANESE ECONOMY AND BUSINESS

日本経済経営研究所

Working Paper Series

September 2007, No. 255

# The Real Effects of Asset Market Bubbles: Loan- and Firm-Level Evidence of a Lending Channel

Jie Gan

This paper is posted at DigitalCommons@Columbia. http://digitalcommons.libraries.columbia.edu/japan

# The Real Effects of Asset Market Bubbles: Loan- and Firm-Level Evidence of a Lending Channel

Jie Gan\*

Hong Kong University of Science and Technology

#### **Abstract**

This paper studies how a shock to the financial health of banks, caused by a decline in the asset markets, affects the real economy. The land-market collapse in Japan provides an ideal testing field in separating the impact of a loan supply shock from demand shocks. I find that banks with greater real estate exposure have to reduce lending. Firms' investment and market valuation are negatively associated with their top lender's real estate exposure. The lending channel is economically important: it accounts for one-third of lending contraction, one-fifth of the decline in investment, and a quarter of value loss.

JEL Classification: G21; C41

Key Words: Bubbles; Bank liquidity shock; Bank health; Bank relationships; Lending channel

Address: Department of Finance, School of Business and Management, Hong Kong University of Science and Technology, Clear Water Bay, Kowloon, Hong Kong. Email: jgan@ust.hk; Tel: 852 2358 7665; Fax: 852 2358 1749.

\* I thank Kalok Chan, Murillo Campello, David Cook, Joseph Fan, Jun-Koo Kang, Yasushi Hamao, Peter Mackay, Stewart Myers, Joe Peek, Eric Rasmusen, David Scharfstein, Jeremy Stein, Bent Vale, Hao Wang, and conference participants at the 2004 Financial Intermediation Research Society Conference, the 2005 RFS-IU Conference on The Causes and Consequences of Recent Financial Market Bubbles, the 2005 City University of Hong Kong Finance Symposium, and the 2005 EFA Conference for helpful comments. I also thank Carol Tse and Crystal Cheung for able research assistance and Virginia Unkefer for excellent editorial assistance. I acknowledge the financial support from the Hong Kong RGC Research Grant (HKUST6280/04H). This paper was previously titled "How Does a Shock to Bank Health Unrelated to Firm Performance Affect Firm Performance?"

#### Introduction

In recent years, there have been increasing concerns among academics and policy makers about the real consequences of large swings in prices of assets, such as equities and property. Banks, the dominant financial institutions in most countries, are an important part of the interaction between the real and financial sectors of the economy. They typically have significant exposure to the asset markets, either through their lending to the real estate sector or, in many countries, through their direct holdings of stocks and land. Therefore, the booms and busts in asset markets are likely to affect the financial conditions of banks. Thus, how shocks to bank health or liquidity, as a result of volatile equity and property prices, influence corporate investment and performance is critical to our understanding of the real effect of "bubbles" (or extreme movements in asset prices).

The objective of this paper is to provide evidence of the real consequences of bubbles by identifying and quantifying the economic impact of an adverse liquidity shock to banks induced by a large decline in the asset market. There are two necessary conditions for such a "lending channel" to work. First, as pointed out by Stein (1998), banks are credit constrained as the asymmetric information problem in the capital market prevents them from issuing uninsured debt and/or equity to offset a shortage of liquidity. Thus a shock to the financial conditions of banks results in reduced lending. Second, as emphasized in the recent banking literature (see James and Smith (2000) for a survey), bank loans are special and banking relationships are valuable. Consequently, when an exogenous shock forces banks to cut back on lending, firms cannot readily substitute other sources of financing. Thus, firms face less available credit, which results in reduced investment and worse performance. These two hypotheses collectively characterize the mechanism of a lending channel through which an adverse bank liquidity shock may affect the real sector. So far, however, there have not been any studies that directly test both of these two hypotheses.

The dramatic collapse in the land market in Japan in the early 1990s provides a natural laboratory to test these two hypotheses. Between 1990 and 1993, there was almost a 50% drop in land prices. Japanese banks historically had significant exposure to the land market, both through their lending to the real estate sector and through their direct holding of land. At the end of 1989, banks on average held 6.4% of their assets in real estate loans and, if marked to market, carried 7.9% of land on their book. This institutional background helps me to overcome an important empirical difficulty in separating the economic impact of a loan supply shock from contemporaneous demand shocks, namely, the endogeneity of bank financial condition to firm

performance. When land prices dropped by half in Japan, a shock that is unambiguously exogenous to any individual bank, banks were hurt differentially according to their exposure to the real estate sector prior to the shock. Thus the collapse of land prices induced an exogenous variation in the financial conditions of banks that is unlikely to be related to firm performance.

More importantly, this setting, along with a unique dataset of matched bank-firm loans, helps to deal with another source of endogeneity -- endogenous selection of bank-firm relationships. For example, the same shocks that trigger banking problems may also reduce firms' investment opportunities and thus their demand for credit. In addition, the adverse shock might hurt the balance sheets of borrowers and their credit worthiness, resulting in lower borrowing capacities. If firms more affected by the adverse shock somehow are selected into banks with more real estate exposure, one would observe a spurious relationship between a shock to bank financial condition and firm performance. This paper deals with this problem by employing a rich dataset of matched loans between banks and firms. These data allow me to exploit the fact that there are firms that borrow from multiple banks with different levels of real estate exposure. I can examine whether the *same* firm borrowing from two different banks would receive less financing from the bank with more real estate exposure.

Two sets of results are presented that characterize how the burst of the bubble in the property market in Japan in the early 1990s is transmitted to firms in the manufacturing sector, whose demand for credit is not directly affected by the shock. I first use the individual loan level data to show that, after fully controlling for borrower characteristics, banks with greater exposure to the real estate sector prior to the shock had to cut back lending. This finding supports the view that banks are credit constrained and that an exogenous shock to bank liquidity leads to reduced lending. The second set of results is at the individual firm level. Japanese industrial firms receive, on average, about one-third of their loans from their largest lenders. I find that the exposure of the top lender to the real estate sector negatively affected firms' fixed investment and stock market valuation, which supports the view that reduced bank financing forces firms to pass up profitable investment opportunities. The lending channel is economically important: the shock from the real estate market accounts for almost one-third of economy-wide contraction in bank lending to the manufacturing sector and accounts for about one-fifth of the decline in fixed investment and a quarter of the drop in stock market valuation. To my knowledge this is the first study that provides large-sample loan-level evidence on how a shock to banks influences firms' investment behavior and stock market performance.<sup>2</sup> Moreover, based on a large sample of publicly traded firms and banks, this paper quantifies the economy-wide impact of a collapse in the property market through the lending channel.

This paper extends a few strands of empirical literature at both the micro and the macro levels. The first is the literature on valuable banking relationships, which is related to the second hypotheses discussed above. Slovin, Sushka, and Polonchek (1993), Gibson (1995), and Kang and Stulz (2000) pioneered the approach of using the effect of bank distress announcements on client firms' performance as evidence of valuable banking relationships. As is discussed in more detail later, however, the results in the subsequent work are mixed (e.g. Bae, Kang, and Lim (2002); Ongena, Smith, and Michalsen (2003)). Related to the first hypothesis in this paper, a strand of literature examines how changes in monetary policy affect bank lending activities, i.e., the "credit channel" (see Bernanke and Gertler (1995) for a survey). The main challenge is to separate a loan supply effect from a loan demand effect. Recent work by Kashyap and Stein (2000) and Campello (2002) addresses this concern by exploiting variations in banks' access to liquidity to identify the credit channel of monetary transmission. The third strand of literature has used aggregate data to examine whether a bank loan supply shock affects economic activity. The results are, again, mixed (e.g., Bernanke (1983); Peek and Rosengren (2000); Ashcraft (2006)). Some of the inconsistencies in the previous studies are likely due to the empirical difficulties discussed above. More importantly, they point to the importance of using disaggregated data at the loan level and the firm level to pin down the mechanism through which shocks to banks affect client firms so that causality can be established.

The paper proceeds as follows. Section 2 describes the natural experiment. Section 3 presents the loan-level evidence on the effect of a shock to banks. Section 4 presents the firm-level evidence on the effect of a shock to banks. Finally, Section 5 presents a conclusion.

# I. The Land-Price Collapse as a Natural Experiment

#### 1.1. Previous Studies on the Effect of Bank Health

Researchers have studied the second hypothesis of a lending channel, i.e., the value of durable banking relationships (see James and Smith (2000) for a survey of this literature). Slovin, Sushka, and Polonchek (1993), Gibson (1995), Kang and Stulz (2000) pioneered the approach of examining the effects of bank distress announcements on client firm performance. Slovin, Sushka,

and Poloncheck (1993) report significantly negative wealth effect of Continental Illinois Bank's de facto failure on client firms. However, these authors focus on only one bank, which limits the applicability of their results. Gibson (1995) and Kang and Stulz (2000), both using a sample of Japanese firms, present evidence that bank health affects firm performance, as reflected in the firm's reduced investment and lower stock market valuation. As pointed out by Bae, Kang, and Lim (2002), however, the interpretation of their results, is complicated by the fact that the bank-firm relationship itself and bank health are not necessarily exogenous to firm performance.

Recent work by Bae, Kang, and Lim (2002) and Ongena, Smith, and Michalsen (2003) is an important step forward in addressing the endogeneity issue. Both use the event-study approach and examine how exogenous shocks to banks affect the equity value of their client firms but find contrasting results. Bae, Kang, and Lim (2002) find that negative news announcements about the Korean banks, including the bankruptcy of a client firm, credit downgrading, deterioration of the Bank for International Settlements (BIS) ratio, etc., have negative effects on the value of their client firms. In contrast, Ongena, Smith, and Michalsen (2003) report that bank distress announcements had only small and temporary impacts on the equity value of Norwegian firms during the Norwegian banking crisis. In addition, Djankov, Jindra, and Klapper (2000) find announcements of bank closure have significantly negative effects on firms in Korea but not in Indonesia and Thailand. The advantage of the event-study approach is that, to the extent that the news announcements are unanticipated, it is possible to interpret the results from the event studies in a causal sense. The event-study approach, however, limits the scope of the analyses and does not provide evidence on the real impact of banking problems.

Related to the first hypothesis of a lending channel, a strand of literature studies how liquidity shocks induced by changes in monetary policy affect bank lending activities, i.e., the "credit channel" (see Bernanke (1995) for a survey). Bernanke and Blinder (1992) and Kashyap, Stein, and Wilcox (1993) present evidence that bank loan volumes decrease during monetary contractions. The latter two papers further document a rise in commercial paper issuance during such times. The main challenge is to identify the impact of a loan supply shock from contemporaneous loan demand-side shocks. Kashyap and Stein (2000) address this challenge by demonstrating that more liquid (large) banks are less likely to reduce lending during monetary contractions than are less liquid (small) banks. In a recent paper, Campello (2002) further exploits the differential responses of small banks with and without multi-bank holding company affiliations.

The evidence suggests that internal capital markets in bank holding companies weakens the impact of monetary policies on bank lending.

Using aggregate data, some studies have examined whether a bank loan supply shock affects economic activity. The results are, again, not conclusive. Using data from the Great Depression, Bernanke (1983) and Calomiris and Mason (2003) find a significant impact of bank loan supply shock on real economic activities. Peek and Rosengren (2000) document the negative effect of the Japanese bank crisis on construction activity in the United States in the 1990s. On the other hand, recent work by Driscoll (2003) and Ashcraft (2006) do not find significant evidence that bank failure / monetary contractions are followed a decline in output.

Some of the inconsistencies in the previous research findings are probably related to the two endogeneity problems discussed in the introduction. This paper seeks to identify a source of exogenous shock to bank financial conditions and, exploiting disaggregate data at both the loan level and firm level, to examine the mechanism through which an adverse shock to bank health affects corporate investment and performance.

#### 1.2. Institutional Background and the Natural Experiment

Land prices in Japan almost tripled in the second half of the 1980s. At its peak in 1990, the market value of all the land in Japan, according to several estimates, was four times the land value of the United States, which is 25 times Japan's size (Cargill, Hutchison, and Ito, 1997). The boom was followed by an equally sharp fall in the early 1990s. Between March 1990 and the end of 1993, the price of land dropped by almost one half. Meanwhile, stock prices in Japan experienced a similar pattern of boom and bust.

Japanese banks had significant exposure to the land market. They actively lent to the real estate sector: at the end of 1989, real estate loans on average constituted 6.4% of the total assets (the median was 5.0%). Banks also held land directly: at the end of 1989, the market value of land as a percentage of total bank assets averaged 7.9% (the median was 4.8%). When the land value dropped by half, many of these real estate loans went bad and banks suffered significant capital losses on the land they held.<sup>3</sup>

This shortage of liquidity was exacerbated by another development in the 1990s, that is, the tightening of capital requirements due to the introduction of the Basel Accord in 1993. During the rest of the 1990s, Japanese banks came under increasing pressure to meet the capital requirements. Although the interest rates were low in Japan in the 1990s, the tightening of capital requirements

means that, as long as capital market imperfections prevent banks from issuing equity, the capital losses in real estate will be translated into less loanable funds.

Such a setting, along with a rich dataset of matched lending between banks and firms, provides an excellent laboratory to examine the transmission mechanism of a shock to banks. It helps to deal with the endogeneity problems commonly encountered in separating the effect of a loan supply from demand-side shocks, namely, the endogeneity of bank financial condition and endogenous selection of banking relationships. Regarding the endogeneity of bank health to firm performance, when land prices dropped by one half, banks were hurt proportionally to their exposure to the real estate sector immediately prior to the shock. Thus, the shock induced an exogenous variation in bank financial condition that is plausibly exogenous to firm performance. Regarding the endogenous selection of bank-firm relationships, since the real estate exposure is not an indication of bank quality ex ante, this setting mitigates the concern of selection of banking relationship by firms. More importantly, with a unique sample of matched lending between borrowers and lenders, I can examine whether the *same* firm borrowing from two different banks would obtain a smaller loan from the bank with more real estate exposure, which controls for the endogenous selection of banking relationships.

One institutional factor that might affect my tests is the liberalization of the bond market in the 1990s. Before 1990, the public debt market was highly regulated and issuing firms had to meet strict accounting criteria. In November 1990, all the official restrictions were lifted. Therefore, it is possible that an observed relationship between banks' real estate exposure and lending may not be due to the weakened financial conditions of banks but due to reduced demand for bank loans as firms rely more on bond financing. This concern, however, should not affect my loan-level lending tests because the matched bank-firm lending data allows one to fully control for (observable or unobservable) firm characteristics by examining how lending to the *same* firm changes when its banks' real estate exposure varies.

Access to the bond market may, however, affect the firm-level tests of investment behavior and stock returns. The public bond market may provide firms with an alternative source of financing to make up for the decline in bank loans, reducing the impact of the lending channel. On the other hand, if access to bond market means loosened banking relationships and if, during an economic downturn, the bond market is not as liquid and thus is not a viable alternative, access to the bond market may even lead to a more severe shortage of financing. Thus the impact of access to the bond market is not ex ante clear cut. As discussed later in the empirical analysis, probably

reflecting the two offsetting effects, access to the bond market does not have significant impact on in firm-level investment and stock returns.<sup>4</sup> Moreover, controlling for bond-market access does not alter the main results about the lending channel.

Overall, the collapse in the Japanese land market provides a unique setting to test both of the two hypotheses related to the transmission mechanism of a shock. In what follows, I first provide evidence on the credit constraints of banks based on the loan level data with matched borrowers and lenders. Then I provide evidence on the real impact of the shock using firm level data.

#### II. Loan Level Evidence on the Effect of a Shock to Banks

In this section I test the first hypothesis on the transmission mechanism of a shock to banks, that is, banks are credit constrained. When they are hit by an exogenous shock, they can not find enough financing to make up for a shock-induced shortfall in funds. In the current setting, the testable hypothesis is that banks with greater exposure to the real estate sector become less healthy (or more liquidity constrained) and therefore are more likely to reduce lending after the shock.

#### 2.1 Empirical Methodology

To identify how banks' financial condition affects bank lending, I first need an exogenous measure of banks' financial condition. As discussed earlier, when land prices dropped by half between 1990 and 1993, an exogenous shock to any individual bank, banks suffered varying degrees of capital losses according to their pre-shock exposure to the real estate sector. Thus the banks' pre-shock real estate exposure serves as an exogenous measure of a bank's financial condition.

Then if firms are randomly assigned to banks, the analysis would be straightforward. I could simply focus on lending behavior of banks with and without much real estate exposure. The difference between the two groups can be attributed to the effect of bank financial condition induced by the land-price collapse. However, firms are not randomly assigned to banks. One can easily construct examples where endogenous selection of banking relationships might drive an observed relationship between the bank's financial condition and firm performance. One such example is that "bad" firms (e.g., those with less investment opportunities) choose to borrow from weak banks to avoid monitoring and discipline. Both these "bad" firms and weak banks are more

vulnerable to external economic shock, leading to a spurious relationship between bank financial condition and firm performance.

In general, an important challenge in separating the effect of a bank loan supply shock is to control for unobserved firm characteristics that might affect loan demand. For example, the same shocks that trigger banking problems may also cause a decline in the investment opportunity of firms and thus the demand for credit by firms. In addition, the adverse shock might hurt the balance sheets of borrowers and their credit worthiness, which leads to reduced lending. If firms more affected by the adverse shock somehow are selected into banks with more real estate exposure, one would observe a spurious relationship between shocks to a bank's financial condition and firm performance.

My empirical strategy is to employ a rich data set of matched lending between lenders and borrowers, which allows one to exploit the fact that there are firms that borrow from multiple banks. Using these firms, I can examine whether the *same* firm received less financing from the unhealthy bank and more from the healthy bank. In a regression framework, the idea is to fully control for the firm characteristics (both observable and unobservable) through the firm fixed effects and therefore isolate the effect of a shock to a bank's financial condition on lending. Specifically, I estimate the following model:

$$Lending_{ij} = a + b RE \ Exposure_i + c \ Bank \ controls_i + d \ Relationship \ characteristics_{ij} + u_j, \tag{1}$$

where subscript i indexes the banks; j indexes the firms; and  $u_i$  is the firm fixed effects.

The dependent variable,  $Lending_{ij}$ , is a measure of lending from bank i to firm j between 1994 and 1998. The availability of credit is measured using the log of loan growth defined as a firm's average long-term borrowing from a particular bank between 1994 and 1998 normalized by the average long-term borrowing from the same bank during the five years prior to the shock (between 1984 and 1989). This focus on long-term lending is because, as pointed out by Hibara (2001), long-term loans are mainly associated with financing of Japanese corporate fixed investment. Since it is difficult to know how long it takes for the real estate losses to show up in lending, I examine the average lending five years after the shock. The bank control variables, which will be introduced shortly, are also averaged across years.

Using the amount of borrowing to measure credit availability implicitly assumes that the amount of debt used is the amount of credit available to the firm. This assumption is defensible for

two reasons. First, after the collapse of stock and land prices, banks, facing mounting non-performing loans, had to tighten credit. On the borrower's side, as most bank loans were backed by land as collateral, the collapse in land prices significantly reduced the firms' ability to collateralize, resulting in tighter credit constraints. The contractual features of long-term loans necessarily mean that the loan balances adjust more slowly than desired by the lender. Therefore, after loan demand is controlled for, a large loan balance can arise both from the lender's willingness to lend and from lending decisions in the past. Normalizing loan balances by the amount borrowed in the earlier years helps separate out the effect of prior lending decisions.

RE Exposure is the bank's exposure to the real estate sector, which is used to measure banks' financial condition. It includes two variables. The first is the bank's real estate loans as a percentage of its total assets, measured in 1989, the year prior to the shock (% Real Estate Loans). The other is the market value of land the bank held at the end of 1989 as a percentage of its total assets (% Land Holding). As only the book value of land is available, I follow Hayashi and Inoue (1991) and use perpetual inventory method to construct the market value of land. The coefficient d captures the effect of the shock to a bank's financial condition on lending and is expected to be negative.

Bank control variables are fairly standard, including bank size, Tobin's q, and liquid assets over total assets. Bank size is measured as the natural log of total assets. Tobin's q, defined as the market-to-book value of the asset, measures a bank's market power or franchise value. The effect of Tobin's q on lending, however, is ex ante unclear. On one hand, to the extent that Tobin's q reflects future investment opportunity, it should be related to more lending. On the other hand, franchise value has been shown to be (negatively) related to banks' incentive to take risk. This is because, with deposit insurance (implicit or explicit), bank owners tend to take excessive risk as their payoff resembles a call option whose value increases with the risk (Keeley (1990) and Gan (2004)). Franchise value is something banks would lose in case of failure and thus constrains risk taking. To the extent that lending represents risks, higher franchise value would be associated to less lending, as is found in the U.S. data by Demsetz, Saidenberg, and Strahan (1996).

The liquidity of the bank's balance sheet is measured as liquid assets over total assets. Kashyap and Stein (2000) find that banks with more liquid assets are less likely to reduce lending during monetary contractions. Liquid assets, however, can also lead to less lending, for two reasons. The first is related to the risk-shifting incentive discussed above. To the extent cash reflects past profitability and may predict future profitability (i.e., franchise value), banks with more cash may

be more conservative in lending. The second reason is related to precautionary saving. Dasgupta and Sengupta (2004) show that, in a multi-period setting, if firms anticipate being credit constrained in the future, an increase in liquid balances may make them more conservative in investment choices. Empirically, Almeida, Campello, and Weisbach (2004) find that firms tend to save more during recessions. Thus the effect of liquid assets on lending is ex ante ambiguous.

I include in the regressions a set of variables reflecting the strength of the lending relationship in the ten years (1984-1993) prior to my sample period. The first dimension of the relationship is its duration. This should be a proxy for the private information the lender has about the firm. Petersen and Rajan (1994) and Berger and Udell (1995) demonstrate explicitly that firms with long-term relationships with banks receive more credit from these banks and pay lower interest rates on loan commitments.

My second measure of the strength of the relationship is a dummy variable indicating whether the bank was the firm's biggest lender at least once from 1984 to 1993. Sharpe (1990) and Rajan (1992) point out that relationship banking also increases the lender's information monopoly and creates "hold-up" costs and that firms can avoid these hold-up costs by establishing relationships with another bank. Japanese firms borrow from multiple banks. During the ten-year period between 1984 and 1993 prior to the sample period, Japanese firms borrowed from 16 banks on average with a median of 14 banks. Japanese firms, however, did not spread their borrowing evenly across all banks. They on average borrowed about one-third from a single institution. Even firms with over 10 lenders concentrated about 20% (the mean is 22% and the median is 20%) of their borrowing with their largest lenders. Therefore, whether or not the bank is the largest lender signifies how close the firm is to the bank.

Another way to mitigate the "hold-up" problem and to ensure bilateral commitment between the lender and the borrower is for the lender to take an equity stake that allows sharing of future surpluses with the borrower, which is a common practice of Japanese banks. Therefore, the third dimension of the relationship is the percentage of equity stake that the bank has in the firm. Lastly, the institutional setting in Japan suggests that main banks obtain additional information on the group-affiliated firms. Therefore, I include a dummy variable indicating whether the bank is the firm's main bank.

In my sample, of all the firm-bank pairs in 1989, about 35% did not have a lending relationship during the after-shock period between 1994 and 1998. To correct for potential survivorship bias, I estimate a selection model using Heckman's (1979) two-stage regression

technique. The first stage is a probit regression on whether the relationship survived; the second stage is an ordinary-least-squares regression on the log of loan growth. To the extent that the credit allocation is a two-step process in which the bank first makes a decision as to whether or not to lend and then makes a decision as to how much to lend, the selection model provides insight to both decisions. In addition, the bank characteristics prior to the shock is more likely to affect the first lending / no-lending decision whereas the contemporaneous bank characteristics tend to affect the second decision more. Therefore, in the first-stage regression, bank controls are measured in 1989, which naturally serve as instruments to help identify the second-stage regression. In all estimations, I report White heteroskedasticity-consistent standard errors. The results are robust to the GLS estimation with bank random effects accounting for dependence of the error terms due to the unobserved effect at the bank level.<sup>6</sup>

#### 2.2 Data

The main data source is Development Bank of Japan (DBJ) database, which contains long-term lending from individual banks to manufacturing firms (Development Bank of Japan Industry Code 1000-1700) and financial information of these manufacturing firms. The data cover all publicly-traded manufacturing firms and their banks. Financial data on banks is from the NIKKEI NEEDs database. I identify each firm's main bank using *Industrial Groupings in Japan: the Anatomy of Keiretsu* by Dodwell Marketing Consultants. Panels A and B of Table 1 present the summary statistics of the relationship-related variables and bank control variables used in the loan-level analysis.

It is worth noting that my sample does not include real estate firms, whose demand for loans is directly affected by the shock from the real estate market. Moreover, it has been reported that Japanese banks "ever-green" their loans by lending to weak firms, many of them real estate firms, to prevent their loans to go bad (e.g., Hoshi and Kashyap, 2004; and Peek and Rosengren, 2005). Consistent with this argument, lending to the real estate sector steadily increased from the early 1990s till 1998. Therefore, excluding real estate firms increase the power of the tests.<sup>8</sup>

# 2.3 Empirical Results

Column (1a) of Table 2 reports the first-stage probit regression results. The dependent variable equals one if the relationship continues after the shock and zero otherwise. The independent variables are measured at the end of 1989. Relationship variables are important in

determining the change in loan renewals. Duration, being the largest lender to the firm, and being the main bank of the firm all increase the chance of loan renewal significantly (at the 1% or 5% level). As expected, the bank's real estate exposure prior to the shock has a significantly negative effect on loan growth: the coefficients on both *% Real Estate Loans* and *% Land Holding* are significantly negative at the 1% level. Probably reflecting the two opposing effects of Tobin's q, its coefficient is insignificant.

Column (1b) of Table 2 reports results from the second-stage OLS regression, which is estimated based on the subgroup of firm-bank observations that have continued lending relationships. Among the relationship variables, being the biggest lender and being the main bank are related to more financing (at the 1% or 5% level). The coefficient of duration, however, is significantly negative. This could be reflecting that some of the loans granted earlier have been paid back. Compared to its effect on the loan renewal decision, a bank's real estate exposure has a somewhat weaker impact on the actual credit allocation. While the coefficient on % Real Estate Loans is significant at the 1% level, the coefficient on % Land Holding is insignificant. The difference between these two exposures to the real estate sector is probably due to the fact that when real estate loans go bad they have a direct impact on the available funds, whereas land holdings, to the extent that they are not marketed to market, is more of a paper loss. The coefficient on liquid assets is significantly negative, which is consistent with the risk-shifting and the precautionarysaving incentives discussed earlier. Note that this result is in contrast to the findings in Kashyap and Stein (2000) that liquid assets weaken the effect of monetary contractions. Thus, while a more liquid balance sheet can buffer small loan supply shocks (e.g., due to changes in monetary policy), the role of liquidity is more complicated when the bank is hit by a large shock.<sup>9</sup>

In columns (2a) and (2b) of Table 2, I further explore how the banks' capital positions affect lending. In particular, if a bank is well capitalized, the impact of losses from its real estate exposure should have a smaller effect on loan growth. Therefore, a dummy variable (*BIND*) is created that indicates if the bank's capital ratio in 1989 was below the 4% capital requirement and let it interact with the two real-estate-exposure variables. <sup>10</sup> In the selection equation in column (2a), although the capital ratio itself is not significant in explaining loan renewal, the interaction between the *BIND* dummy and % *Real Estate Loans* is significantly negative at the 10% level. In column (2b), the coefficient on the capital ratio is significantly positive (at the 5% level); the coefficients on the interaction terms are not significant. Other coefficients on the bank's real estate exposure remain qualitatively unchanged.

# 2.4 Robustness: The Effect of Bank Specialization

An alternative interpretation of the above results is related to bank specialization.<sup>11</sup> Suppose firms have different types of loans served by different banks. For example, a firm may use Bank 1 for its real estate-related loans and Bank 2 for, say, capital expenditure. Then with a fall in the real estate market and thus real estate-related activity, there is less demand for real estate loans, leading to reduced lending by Bank 1 with more observed "real estate exposure."

This argument is particularly relevant for real estate firms, since their demand for credit is closely tied to the real estate market. This concern, however, is greatly mitigated by the fact that the sample contains only manufacturing firms, not real estate firms. For manufacturing firms, real estate is an input into the production process (e.g., plants and structures) and should mostly be part of their fixed investments. Thus even if firms fund their real estate from one particular bank, as long as real estate borrowing is proportional to the total borrowing, the change in demand for loans from this "real estate bank" will not differ from other banks.

The concern about specialized real estate lending is further mitigated by the fact that the Japanese banks are large and therefore less likely to be specialized (which is typical of a "bank-centered" economy.) During the sample period, their average size is, if converted to U.S. dollars, approximately 80 billion. The smallest bank in the sample has assets of approximately 1 billion U.S. dollars. The Japanese banks are required to report their loans to different sectors (e.g., manufacturing loans, real estate loans, loans to individuals, etc.) I thus calculate the concentration ratio of the largest loan category, defined as the loan amount in the largest category as a percentage of total loans. The maximum concentration ratio is slightly above two-thirds (36%), with the 90 percentile concentration ratio being slightly above a quarter (28%). Moreover, only 8% of banks have real estate loans as the largest category and among these banks, the real estate loans, on average, account for 25% of total lending. Therefore, it is unlikely that small and specialized real estate banks are driving the results.

The following test was performed to further confirm the robustness of the results. If specialized real estate lending to the industrial firms - not bank financial condition - is driving the results, then, if a subset of banking relationships could be identified that are not likely to be driven by specialized real estate lending, the earlier results would not hold for these relationships. One such relationship is that between a firm and its top lender. As discussed earlier, Japanese firms source one-third of their borrowing from the top lender. Since real estate activities unrelated to

fixed capital expenditure must be a small proportion of their overall borrowing, borrowing from the top lender should not be specialized real estate loans. To implement this test, I create an interaction term between the dummy for top lenders and the lender's real estate exposure. If the specialized real estate lending hypothesis is true, one would expect the coefficient on the interaction term to be positive so that the sum of the coefficients of real estate exposure and the interaction term is not significantly negative. The results are displayed in columns (2c) of Table 2. The coefficient of the interaction term is not positive, indicating that specialized real estate lending does not explain the earlier results about the effect of bank financial condition. Interestingly, the interaction term is significantly negative, implying that, although the top lenders generally lend more, the marginal benefit firms get from their top lenders decreases as the top lenders have more real estate exposure and therefore become financially less healthy. This result therefore strengthens the basic finding that banks' financial conditions matter.

While bank specialization per se does not seem to affect the tests, one may be concerned with some pre-sorting of lending relationships that cannot be completely controlled for by firm fixed effects. For example, some banks may have a comparative advantage in funding certain types of loans within industrial loans, e.g., operating funds as opposed to non-operating funds (equipment funds). If this type of loan is somehow "out of favor" after the shock and if this less favored loan type is somehow related to banks' real estate exposure, the earlier results could be driven by a systematic switch of loan types rather than by a bank's financial condition. Although it is not obvious why for large and unspecialized banks, their engagement in certain industrial loan types are systematically related to their lending to the real estate sector or their land holding, it is worthwhile to check for robustness if relevant data are available. Japanese banks report their lending as operating funds versus non-operating (equipment) funds. As a robustness check, the percentage of operating funds prior to the shock (a bank level variable) is included in the estimation. The results are reported in column (2d) of Table 2. 12 It turns out that operating funds have a negative impact on overall lending. This suggests that, when faced with tighter credit, firms economize on their operating funds (e.g., by cutting down on general office expense or advertising spending) before curtailing investment in productive capacity. The coefficients on the real estate exposure remain unchanged, implying that, while the shock might affect the loan demand mix, this phenomenon in itself does not drive the observed effect of a bank's financial condition.

# 2.5 Discussions: The Impact of the Contemporaneous Stock Market Collapse

While it is prohibited in some countries, in Japan, banks typically hold equity in their client firms. It is possible that bank losses in security holdings lead to worsened financial conditions in a similar way as their real estate exposure. Therefore, to gain a complete picture of the real effects of asset market bubbles, I explore the impact of the bank's stock market exposure.

There are, however, several factors that make it unclear how banks' equity holdings might matter empirically. First, as market value accounting is only a recent concept, the "paper losses" from equity holdings may not affect bank liquidity and capital position. Second, banks' equity holdings are typically long -- many equity stakes were purchased as early as the 1950s (Hoshi and Kashyap (2001)). With historical cost accounting, the holding period return is not negative even with the bursting of the bubble. 13 In such cases, banks may sell their stock holdings in an effort to strengthen their capital positions, since the capital gain can count as capital under "revaluation reserves" in the Basle Accord capital requirements. Third, while equity holdings represent the bank's stock market exposure, it may also measure the strength of its client relationships, especially with its larger clients. In identifying the real impact at the firm level, I focus on the firm's most important lending relationship -- the one with its top lenders. Thus my ability to identify a real impact depends critically on how banks cut back lending across firms. For example, if banks cut back lending in equal yen amounts across firms, I may not be able to detect a real impact since there is not enough cross-firm variation in reduction of loans and thus investments. 14 This is all the more likely to happen when loan cutbacks are due to losses in banks' equity holdings: banks with larger equity holdings may be more protective of their large and more dependent borrowers, resulting in an seemingly insignificant real impact on firms. Lastly, empirically quantifying this effect is made difficult by the fact that only the book value of share holdings is reported, which is a noisy measure of banks' losses in the stock market.

In columns (3a) and (3c) of Table 2, I add as an independent variable is added for the bank's stock market exposure, measured as its equity holding as a percentage of total assets prior to the shock in 1989 (% Stock Holdings). I also control for whether the bank sold their equity holdings between 1989 and 1993 (SOLD). While the bank's equity holdings do not significantly affect the decision to renew loans, they are associated with significantly less lending overall (at the 1% level). The coefficient on the SOLD dummy is statistically insignificant. This is probably because banks may have sold their equity holdings for different reasons. On the one hand, banks might sell because they are desperate to come up with capital to meet the regulatory requirements, in which case SOLD is likely to be related to more cutbacks in lending. On the other hand, banks might have

accumulated a larger capital gain, which makes selling a cheaper alternative to other ways of raising capital, in which case *SOLD* should be related to a smaller reduction in lending. Interestingly, the coefficient on bank's land holding now becomes significantly negative. This appears to be related to the fact that banks with greater land holding are more likely to sell their equity holdings to make up for the capital loss, reducing the impact of the bank's land holding in the earlier estimation. Indeed, those banks that sold their equity holdings on average have land holding of 25%, as compared to the 6% land holding among those that did not sell.

In sum, the banks' real estate exposure prior to the shock negatively affects lending in two important ways: it not only lowers the chance of loans being renewed but also reduces the amount of credit actually being granted. A question naturally arises: does such reduction in loans affect the client firms' investment behavior? This issue, as well as the real impact of the bank's stock market exposure, is examined in the next section.

# III. Firm-Level Evidence on the Effect of a Shock to Banks

With firm-level data, this section examines the second hypothesis related to the transmission of the shock to banks, that is, reduced lending results in less fixed investment and worse firm performance. As discussed earlier, Japanese firms borrow close to one-third of their loans from their largest lenders. When the top lender is hit by the real estate shock, if the firm can not substitute other sources of financing for funds that are previously provided by the top lender, the firm would have to cut back its investment. To the extent that firms have to pass by positive-NPV projects, their market valuation will be lowered. Therefore, in what follows, I investigate whether the pre-shock real estate exposure of the firm's top lender is related to the firm's fixed investment and stock market returns in the period subsequent to the land-price collapse.

# 3.1. Top Lender's Real Estate Exposure and Firm Investment

# 3.1.1 Empirical Specification

To examine whether the top lender's real estate exposure negatively affects the firm's ability to invest, I estimate the following baseline model:

The dependent variable, I/K, is measured as the average investment rate for the period 1994-1998, reflecting, again, the fact that it is difficult to know how long it takes for the shock to take effect. Accordingly, the firm-level variables are also averaged across years unless otherwise specified. The main variables of concern are the top lender's real estate exposure, measured as the real estate loans of the firm's top lender (% Re al Estate Loans Top) and the top lender's land holding (% Land Holding Top). Since my loan-level analysis suggests that the bank's capital position matters, I also control for the bank's capital ratio in 1989. The coefficients d and e are expected to be negative.

Firm controls include the usual set of variables in an investment equation: I have Tobin's q to measure investment opportunity, the cash flow and the cash reserve normalized by the capital stock to measure internal liquidity, and industry dummies to control for industry-wide shocks. In a recent study, Gan (2006) finds that fixed investments by Japanese firms were negatively related to their land holding prior to the shock, a measure of collateral losses they suffered. I therefore control for the firm's land holding in 1989, defined as market value of land normalized by the capital stock. I also control for the timing of the land purchase, measured by the proportion of land (in market value) purchased between 1988 and 1990, the period of rapid land price increase.

Lang, Ofek, and Stulz (1996) find that future growth and investment are negatively related to leverage, particularly for firms with high debt ratios. In the current setting, the effect of firms' collateral losses may also depends on firm leverage, with highly leveraged firms investing less due to more binding borrowing constraints. I therefore control for leverage and also create a dummy variable indicating leverage above the median and let it interact with firm land holding. Panel C of Table 1 presents a summary of the variables used in the firm-level analysis.

Regarding the estimation technique, note that the investment rate is right-skewed, with the median being about one-third of the mean. When I estimate the investment equation using an ordinary-least-squares (OLS) regression, I find that the distribution of the residual is still skewed, suggesting that the OLS estimators are not efficient. Therefore, I estimate the median or least absolute distance (LAD) regressions. LAD minimizes that sum of the absolute deviations rather the sum of the squared deviations. Therefore, it is less sensitive to the tail of the distribution or to outliers. Additionally, since for skewed data the median is generally a more efficient measure of the

center of the data than does the mean, the precision of the estimates will also increase. <sup>15</sup> The standard errors are calculated based on the method suggested by Koenker and Bassett (1982).

#### 3.1.2 Data

The firm-level data mainly come from the tapes compiled by the Development Bank of Japan (DBJ). The DBJ database contains detailed accounting data on all non-financial firms listed on various stock exchanges from 1956 onwards. For investment analyses, the DBJ database has several advantages over the NIKKEI NEEDS database, a popular database used in Japanese economic studies. It provides a detailed breakdown of five depreciable capital goods, as well as asset-specific gross and current period depreciation, which together enable a more accurate calculation of the replacement cost of capital and the investment rate net of asset sale. This is why Japanese data have been used frequently in tests of investment models (e.g., Hayashi and Inoue 1991). The stock market data are from the NIKKEI NEEDS database.

The sample contains all manufacturing firms in the DBJ database for which stock market data are available for the period 1986-98 and information on the top lender's real estate exposure is available for 1989. I drop from the analysis firms that do not have enough data to construct a capital-stock measure or have missing stock price data, firms that were involved in mergers and acquisitions between 1989 and 1998, and, if a firm changes its accounting period, the year in which such a change occurs. The Japanese fiscal year ends in March. However, many firms file late in the year. I follow the standard approach and define the fiscal year for a particular observation as the previous year if the firm filed before or in June, and as the current year if the firm files after June. The final sample contains 420 firms. <sup>16</sup>

# 3.1.3 Empirical Results

Column (1) of Table 3 shows that the real estate exposure of the firm's top lender, as measured by real estate loans, has a significantly negative impact on fixed investment (at the 5% level). The coefficient on land holding of the top lender is statistically insignificant. As is discussed below, however, the coefficient is significant for the less capitalized lenders. The top lender's capital ratio prior to the shock is positively associated with the investment rate of the borrowing firm (at the 10% level), suggesting that all else equal, better capitalized banks are able to lend more.

Consistent with Gan (2006), the firm's land holding prior to the shock is related to a lower investment rate (significant at the 1% level). The coefficient on the interaction term between the

high leverage dummy and firm's land holding is, as expected, significantly negative (at the 1% level). To the extent that land serves as collateral for corporate borrowing, the shock also worsens the borrower's balance sheet position and thus their ability to borrow, especially when they are already high leverage. These results suggest a collateral channel a la Bernanke and Gertler (1990) and Kiyotaki and Moore (1997) and strengthen the argument that, in a test of the impact of the financial conditions of banks, it is important to control for the borrower's "balance-sheet effect". Lastly, the coefficient on leverage is significantly positive at the 10% level, probably because in a bank-dominated economy and at a time of severe credit rationing, how much firms can borrow also proxy for firm quality and/or lending relationships. However, leverage does hurt growth in the sense that it worsens the impact of collateral losses.

In column (2) of Table 3, I control for firm's access to alternative sources of debt financing, i.e., its access to the public bond markets. As pointed out by Hoshi and Kashyap (2001), after the liberalization of the bond market, Japanese firms need at least an investment grade (i.e., a rating equivalent to S&P's BBB rating or higher) to issue public debt. I code a dummy variable indicating whether or not a firm meets these criteria at least four years out of the five-year period during 1994-1998 and include it in the estimation. <sup>17</sup> Consistent with the earlier discussion that the effect of access to the bond markets is not ex ante clear-cut, the coefficient on access to the bond markets is not statistically significant.

The impact of the top lender's pre-shock exposure to the real estate sector may depend on how concentrated the firm's borrowing is on this lender. The more the firm relies on its top lender, the more it will be hurt when this lender has to cut back lending. Therefore, in column (3) of Table 3, I control for the concentration of borrowing from the top lender, defined as the top lender's share in the firm's total long-term loans outstanding. I also create a dummy variable indicating if the concentration is above the median and let it interact with the two measures of real estate exposure. Note that this specification also addresses a potential alternative explanation of the earlier results: if firms happen to concentrate their borrowings more on top lenders with greater real estate exposure, perhaps because these banks had more funds to spare in the late 1980s, the earlier findings may merely reflect such concentration in borrowing rather than the effect of the top lender's real estate exposure. As shown in column (3) of Table 3, the concentration of loans from the top lender does have a negative impact on firm investments (at the 1% level). However, neither of the two interaction terms is significant and, by an F-test, I cannot reject the hypothesis that they are jointly

insignificant in explaining firm investment (p-value = 0.3115). On the other hand, the results on the real estate exposure remain unchanged.

Next I examine how the effect of the top lenders' real exposure depends on the bank's capital position. In column (4) of Table 3, I create a dummy variable, *BIND*, indicating whether or not the bank had a capital ratio lower than the 4% requirement in 1989 and let it interact with the two measures of real estate exposure. The interaction term between the *BIND* dummy and the top lender's land holding is significantly negative (at the 10% level), suggesting that the insignificant coefficient on % *Land Holding* earlier is driven by those better capitalized lenders. When the top lender is not well capitalized, its exposure due to direct land holding hurts a client firm's ability to invest. On the other hand, the interaction term between the *BIND* dummy and % *Real Estate Loans* is not statistically significant, suggesting that the negative impact of exposure due to real estate lending does not vary significantly across lenders with different capital positions. The difference between the two coefficients on real estate exposures is similar to that in the loan-level tests, consistent with the earlier conjecture that losses from real estate loans dry up funds more directly and therefore are more related to the bank's ability to grant loans.

Lastly, I explore the real effect of the top lender's stock market exposure. In column (5) of Table 3, I control, similar to Table 2, the top lender's equity holdings (% Stock Holdings) and a dummy variable indicating whether the bank sold its equity holdings (SOLD). Neither of the two variables is statistically significant. Consistent with my earlier conjecture that equity holdings reflect the bank's relationships with their clients, the bank seems to be reluctant to cut back loans to those who are dependent on them. To further explore this possibility, I include in the lending equation an interaction term between the top-lender dummy and the bank's equity holdings, the coefficient is positive but not statistically significant (unreported). Thus, unlike the bank's real estate exposure, a significant number of banks, when faced with stock market losses, tend to cut proportionally less loans to more dependent borrowers and thus squeeze smaller borrowers more. Therefore, even if there is a real impact of the lending shock, such an impact is hard to be detected across firms. Another reason for the insignificant stock market effect may simply be that the book value of equity holding is a noisy indicator of the bank's true stock market exposure and thus reduces the statistically power of my tests. These difficulties further highlight the advantage of focusing on the bank liquidity shock arising from the burst of real estate bubble.

Hayashi and Prescott (2002) argue that Japanese firms do not seem to be financially constrained as they hold much more cash than the U.S. firms and small firms have steadily

increased their cash holding since 1996. They point out a slowdown in total factor productivity growth as the main reason for Japan's "lost decade of growth." Cash holding, however, is endogenous. Recent empirical evidence that financially constrained firms save a bigger fraction of their cash balances in recessions is supportive of this view (Almeida, Campello, and Weisbach, 2004). The results in this paper, however, are not inconsistent with Hayashi and Prescott's view. As the shock worsens the financial condition of the banking sector and therefore reduces their ability to lend, to the extent that firms have to cut back on investment in technologies that improves productivity, the total factor productivity in the economy would be reduced.

The literature on the role of banks suggests that a liquidity shock to the banking sector causes firms to underinvestment. However, in the current setting, it is possible that firms took advantage of the increased collateral value due to the land price run-up in the 1980s to borrow excessively and wasted the funds on unprofitable projects. The reduced investments after the collapse simply represent a correction to the overinvestment problem. Although this hypothesis may change the interpretation of the results, it does not refute the effect of declined financial health of banks on firm investments. Nevertheless, this issue is important because it relates to our understanding of macroeconomic impact of bank health. I distinguish between these two hypotheses by examining the firms' stock market valuation. To the extent that firms under-invest and pass up profitable investment projects, their stock market valuation should be lowered.

# 3.2. Top Lender's Real Estate Exposure and Firm Value

# 3.2.1. Empirical Specification

To examine whether the stock market valuation of firms is negatively related to their top lenders' real estate prior to the shock, I estimate the following model:

$$RET_{j} = a + b \% Real Estate Loans_{j}^{Top} + c \% Land Holding_{j}^{Top} + d Firm Controls_{j} + \varepsilon_{j}, \qquad (3)$$

where the dependent variable is the buy-and-hold return for the period 1994-98 (RET). The main variables of concern are the top lender's real estate exposure: the top lender's real estate loans (% Re al Estate Loans $_j^{Top}$ ) and the land holding of the firm's top lender (% Land Holding $_j^{Top}$ ). The coefficients b and c are expected to be negative. Similar to Table 3, I control for the bank's capital ratio in 1989.

In estimating Equation (3), I consider, in addition to the effect of bank financial condition, several non-mutually exclusive hypotheses similar to those in Kang and Stulz (2000) that may explain the cross-sectional returns for the period subsequent to the land-price collapse. The first is a "bubble-size" explanation, which implies that firms that enjoyed the boom during 1986-89 had a larger bubble in its valuation and would perform worse later. <sup>19</sup> Therefore, I include the return during 1986-89 as a control variable (RET8689). To further distinguish this hypothesis from the Capital Asset Pricing Model (CAPM), I also control for the beta (BETA8689), which is calculated from the monthly return during 1986-89. Another explanation is that, concurrent with the shock to land price, there is a shock to the profitability of projects undertaken in the second half of the 1980s so that these projects ceased to be profitable. Therefore, I control for a forecast of cash flow, using the cumulative cash flow during 1986-89 divided by the firm's total assets. According to this hypothesis, the coefficient on this variable should be negative.

Similar to the investment equation, I control for the firm's land holding prior to the shock  $(Land/K^{1989})$  and the timing of land purchases (% Recent Purchase). I also control for two valuation variables observed in 1989, namely, the price-earnings ratio and the book-to-market ratio. Lastly, as the dependent variable is the equity return, which is a function of leverage for any given return on assets, I control for leverage at the beginning of the sample period in 1993. Panel C of Table 1 presents a summary of the variables used in the analysis.

# 3.2.2. Empirical Results

As shown in Table 1 (Panel C), between 1994 and 1998, the sample of firms experienced an average loss in equity value of 26% and a median loss in value of 34%. Column (1) of Table 4 shows that the firm value is negatively related to the top lender's land holding (significant at the 5% level). The coefficient on real estate loans is not statistically significant. However, as is discussed later, this is related to the capital position of the bank. Supportive of the bubble-size hypothesis, column (1) of Table 4 shows that the greater the gain prior to the shock between 1986 and 1989, the greater the fall in the post-shock period. Leverage is significantly positive, probably suggesting that firms that can secure borrowing are better firms and/or those with better relationships. Consistent with the earlier finding that firms with land holdings prior to the shock suffered greater losses of collateral and therefore faced greater credit constraints, the coefficient on the firm's land holding in 1989 is significantly negative (at the 10% level).

Column (2) of Table 4 includes some additional controls. As share prices fell more than the fundamentals, firms with more security holdings are expected to lose more value to the extent that the security holdings are shares. I control for export sales because given the recession at home, firms that export more may have been less affected. I also control for firms' access to the bond market. None of these additional controls is significant; the earlier results regarding the impact of the top lender's real estate exposure remain unchanged.

In column (3) of Table 4, I control for the firm's concentration of borrowing from its top lender. Similar to the investment equation, the top lender's loan share is negatively related to firm value. However, none of the two interaction terms is significant and they are jointly insignificant based on the F-test (p-value = 0.4952). In column (4) of Table 4, I further examine how the effect of real estate exposure may vary with the top lender's capital position prior to the shock. The interaction term between the *BIND* dummy and the top lender's real estate loans becomes negative and marginally significant at 10% by a one-sided test. Given that the predictions about the real estate exposure are one-sided, this finding indicates (albeit weakly) that although the market does not "penalize" a firm when its better capitalized top lender lends to the real estate sector, if its top lender is not well capitalized, the lender's real estate lending hurts the firm's valuation. On the other hand, results regarding the top lender's land holding remain unchanged. The difference in the two measures of real estate exposure could be reflecting that land holding is more visible than lending to the real estate sector and the market valuation is thus more based on land holding.

The impact of the stock market exposure is presented in column (5) of Table 4. Consistent with the results in Table 3, the stock market related variables are not statistically significant. This finding further supports the earlier argument that, given the data limitations, it is hard to empirically quantify the real impact of the bank's stock market exposure.

As a summary, the results in this section demonstrate that the real estate exposure of a firm's top lender affects the firm's ability to make fixed investments and reduces its stock valuation. These findings, combined with the earlier evidence at the loan level, suggest that a shock to the financial condition of the banking sector is transmitted to the real sector by reducing the bank's ability to grant loans and the firm's ability to invest profitable projects. Are these results economically significant? This question is examined in the next section.

# IV. Economic Significance

I compare the lending behavior of two hypothetical banks that are otherwise identical except that one has real estate exposure at the 25<sup>th</sup> percentile in both measures (% Real Estate Loans = 3.6% and % Land Holding = 2.1%) and the other at the 75<sup>th</sup> percentile (% Real Estate Loans = 7.9% and % Land Holding = 7.8%). Note that the coefficient estimates in column (1a) of Table 1 reflect the marginal impact of change in real estate exposure on the probit index, not probability. Therefore I first calculate the marginal impact of real estate exposure on the probability of relationship continuation evaluated at the mean. They are -0.944 for % Real Estate Loans and -0.879 for % Land Holding. Thus the probability of continued relationship is 4.1 percentage points (=-0.944\*(7.9%-3.6%) lower for the 75<sup>th</sup> percentile bank due to its excess real estate loans and 5.0% percentage points (=-0.879\*(7.8%-2.1%)) lower due to its excess land holding. These effects are moderate compared to the unconditional probability of relationship survival (65%). As to the overall impact of real estate exposure on the loan growth, from column (1b) of Table 2, the 75<sup>th</sup> percentile bank is expected to be have a growth rate over pre-shock level that is 9.2 percentage points (=-2.318\*(7.9%-3.6%) lower than that is for the 25<sup>th</sup> bank. These magnitudes are clearly economically important.

The estimates in Table 3 and 4 are also consistent with a significant effect of the lending shock on fixed investment and firm value. Let's base the calculation on the differential behavior of two otherwise identical firms with top lenders' real estate exposure at the  $25^{th}$  percentile and the  $75^{th}$  percentile. According to column (1) of Table 3, the firm with a top lender who held real estate loans at the  $75^{th}$  percentile would have an investment rate 2.6 percentage points lower than the other firm. If the top lenders face binding capital requirement, firm investment will be further reduced by 6.8 percentage points due to the lender's land holding. The stock market appears to have taken this into account: the firm with the top lender with  $75^{th}$  percentile exposure has a market valuation that is 8.6 percentage points lower due to its lenders' excessive land holding and in the case where the top lender is facing binding capital requirement, the market valuation is 5.2 percentage lower due to the lender's excessive lending to the real estate sector. These effects are economically significant.

The above calculations only compare banks at extremes of the spectrum of real estate exposure. To gauge the total impact of the shock, I integrate over the distribution of the real estate exposure. In particular, I take the actual real estate exposure measures and assume that for banks with these measures above the  $25^{th}$  percentile the credit constraints are binding. For example, to

obtain the aggregate impact of the shock on the loan growth, I take the estimate of -2.138 in column (1) of Table 2 (the most conservative estimate) and calculate an aggregate of the incremental impact of % Real Estate Loans above the 25<sup>th</sup> percentile across bank-firm pairs. It turns out the total loan growth is 3.5 percentage points lower than it would be if all banks were unconstrained. The total fixed investment is lowered by 1.1 percentage point due to banks' real estate lending and 0.16 percentile due to their land holding. Aggregate firm value is lowered by 2.8 percentage points due to banks' land holding and by 0.7 percentage point due to banks' real estate lending. During the sample period, firms in the sample lost 13.8% in total value. Thus the shock to banks accounts for about a quarter (25.2%) of such a loss.

It is worth noting that the results above are based on publicly traded banks and firms, which represent perhaps the most important component of the economy. For smaller and younger firms, the effect can be arguably even more pronounced as they are more affected by the information asymmetry in the credit market due to higher "information costs." Therefore, the above economic magnitude can be considered as a lower bound of the real impact of the shock.

# V. Conclusion

Exploiting a unique economic setting and a unique data set, this paper identifies and quantifies the full economic impact of a powerful lending channel that transmits downward movements in asset markets to the real economy. It provides two levels of evidence characterizing such a transmission mechanism. At the individual loan level, I find that banks are credit constrained: when they are hit by the shock in the land market, banks with greater real estate exposure prior to the shock have to cut back lending in the five years subsequent to the shock. At the individual firm level, I find that, consistent with the view that firms cannot easily substitute other sources of financing for bank credit, the borrowing firms' behavior is significantly affected by their top lender's real estate exposure: they invest less and lose market value. To my knowledge, this is the first paper providing large-sample loan-level evidence on the transmission of a banking shock to firms' real (investment) decisions and market valuation. Moreover, it is also the first paper that is based on all publicly traded firms in an economy and thus quantifies the full economic impact of a loan supply shock.

While this paper emphasizes a strong lending channel that transmits an asset market bubble into the real economy, Gan (2006) documents another channel through which the real and financial sectors of the economy interact. That is, the burst of asset market bubbles is likely to impair collateral value and thus diminish firms' to borrower to finance investment projects, because most bank loans are collateralized and land is an important form of collateral around the world. This collateral channel, together with the lending channel identified in this paper, points to two powerful ways by which asset market bubbles affect the real economy.

Japan's experience has important implications for the U.S. and other countries. Around the world, the property markets have enjoyed unprecedented price increases in recent years. As a result, banks have enjoyed large increases in real estate related lending business. There has been hot debate on what would happen to the economy if there were a "correction" in the property markets. The Japanese experience in the 1990s suggests that a decline in the property market may have a large adverse impact on the real economy through the interaction between the asset markets and the credit markets.

#### References

Almeida, H., M. Campello, and M. Weisbach, 2004, "The Cash Flow Sensitivity of Cash," *Journal of Finance*, 59, 1777-1804.

Ashcraft, A., 2006, "New Evidence On The Lending Channel," *Journal of Money, Credit, and Banking*, 38, 751-775.

Bae, K. H., J. K. Kang, and C. W. Lim, 2002, "The Value of Durable Bank Relationships: Evidence From Korean Banking Shocks," *Journal of Financial Economics*, 64, 181-124.

Berger, A., and G. Udell, 1995, "Relationship Lending and Lines of Credit in Small Firm Finance," *Journal of Business*, 68, 351-381.

Bernanke, B. S., 1983, "Nonmonetary Effects of the Financial Crisis in the Propagation of the Great Depression," *American Economic Review*, 73, 257-276.

Bernanke, B. S., and A. S. Blinder, 1992, "Money, Credit and Aggregate Demand," *American Economic Review*, 82, 901-921.

Bernanke, B. S., and M. Gertler, 1990, "Financial Fragility and Economic Performance," *Quarterly Journal of Economics*, 105, 87-114.

Bernanke, B. S., and M. Gertler, 1995, "Inside the Black Box: The Credit Channel of Monetary Policy Transmission," *Journal of Economics Perspective*, 9, 27-48.

Calomiris, C. W., and J. R. Mason, 2003, "Consequences of Bank Distress During The Great Depression," *American Economic Review*, 93, 937-947.

Cargill, T. F., M. Hutchison, and T. Ito, 1997, "The Political Economy of Japanese Monetary Policy," Cambridge and London: MIT Press.

Campello, M., 2002, "Internal Capital Markets in Financial Conglomerates: Evidence From Small Bank Responses to Monetary Policy," *Journal of Finance*, 57, 2773-2805.

Djankov, S., J. Jindra, and L. Klapper, 2000, "Corporate Valuatin and The Resolution of Bank Insolvency in East Asia," World Bank Working Papper.

Dasgupta, S., and K. Sengupta, 2004, "Financial Constraints, Hurdle Rates, and Investment Activity: Implication from a Multi-period Model," Hong Kong University of Science and Technology Working Paper.

Demsetz, R.S., M. R. Saidenberand, and P. E. Strahan, 1996, "Banks With Something To Lose: The Disciplinary Role of Franchise Value," *Federal Reserve Bank of New York Economic Policy Review*, 1-10.

Driscoll, J., 2003, "Does Bank Lending Affect Output: Evidence From U.S. States," FEDS Working Paper 2003-31, Board of Governors of the Federal Reserve.

Gan, J., 2004, "Banking Market Structure and Financial Stability: Evidence from the Texas Real Estate Crisis in the 1980s," *Journal of Financial Economics*, 73, 567-601.

Gan, J., 2006, "Collateral, Debt Capacity, and Corporate Investment: Evidence from A Natural Experiment," *Journal of Financial Economics*, forthcoming.

Gibson, M., 1995, "Can Bank Health Affect Investment? Evidence from Japan," *Journal of Business* 68, 281-308.

Hayashi, F., and E. Prescott, 2002, "The 1990s in Japan: A Lost Decade," *Review of Economic Dynamics*, 5, 206-236.

Hayashi, F., and T. Inoue, 1991, "The Relation Between Firm Growth and Q with Multiple Capital Goods: Theory and Evidence From Panel Data on Japanese Firms," *Econometrica*, 59: 731-53.

Heckman, J., 1979, "Sample Selection Bias as a Specification Error," *Econometrica*, 47, 153-161.

Hibara, N., 2001, "What Happens in Banking Crisis? Moral hazard v. credit crunch," Columbia University Dissertation.

Hoshi, T., and A. Kashyap, 2001, Corporate Financing and Governance in Japan, (MIT Press, Cambridge, MA).

Hoshi, T., and A. Kashyap, 2004, "Japan's Financial Crisis and Economic Stagnation," *Journal of Economic Perspectives*, 18, 3-26.

James, C., and D. Smith, 2000, "Are Banks Still Special? New Evidence on Their Role in the Capital-Raising Process," *Journal of Applied Corporate Finance*, 13, 52-63.

Kang, J., and R. Stulz, 2000, "Do Banking Shocks Affect Borrowing Firm Performance? An Analysis of the Japanese Experience," *Journal of Business*, 73, 1-23.

Kashyap, A., J. C. Stein, and D. Wilcox, 1993, "Monetary Policy and Credit Conditions: Evidence From the Composition of External Finance," *American Economic Review*, 83, 78-98.

Kashyap, A., and J. C. Stein, 2000, "What Do a Million Observations on Banks Say About the Transmission of Monetary Policy?," *American Economic Review*, 90, 407-428.

Keeley, M. C., 1990, "Deposit Insurance, Risk, and Market Power in Banking," *American Economic Review*, 80, 1183-1199.

Kiyotaki, N., and J. Moore, 1997, "Credit Cycles," Journal of Political Economy, 105, 211-248.

Koenker, R., and G. Bassett, Jr., 1982, "Robust Tests for Heteroscedasticity Based on Regression Quantiles," *Econometrica*, 50, 43-61.

Koenker, R., and G. Bassett, Jr., 1978, "Regression Quantiles," *Econometrica*, 46, 33-50.

Koenker, R., and K. Hallock, 2001, "Quantile Regression," *Journal of Economic Perspectives*, 15, 143-156.

Lang, L., E. Ofek, and R. Stulz, 1996, "Leverage, Investment, and Firm Growth," *Journal of Financial Economics*, 40, 3-29.

Ongena, S., D. Smith, and D. Michalsen, 2003, "Firms and Their Distressed Banks: Lessons From the Norwegian Banking Crisis," *Journal of Financial Economics*, 67, 81-111.

Peek, J., and E. Rosengren, 2000, "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States," *American Economic Review*, 90, 30-45.

Peek, J., and E. Rosengren, 2005, "Unnatural Selection: Perverse Incentives and the Misallocation of Credit in Japan," *American Economic Review*, 95, 1144-1166.

Petersen, M., and R. Rajan, 1994, "The Benefits of Lending Relationships: Evidence From Small Business Data," *Journal of Finance*, 49, 3-37.

Rajan, R., 1992, "Insiders and Outsiders: The Choice Between Informed and Arm's Length Debt," *Journal of Finance*, 47, 1367-1400.

Sharpe, S., 1990, "Asymmetric Information, Bank Lending and Implicit Contracts: A Stylized Model of Customer Relationships," *Journal of Finance*, 45, 1069-1087.

Slovin, M., M. Sushka, and J. Poloncheck, 1993, "The Value of Bank Durability: Borrowers As Bank Stake Holders," *Journal of Finance*, 48, 289-302.

Stein, J. C., 1998, "An Adverse-Selection Model of Bank Asset and Liability Management with Implications for the Transmission of Monetary Policy," *RAND Journal of Economics*, 29, 466-486.

<sup>1</sup> Researchers have used aggregate data to study the relationship between bank loan supply and economic activity. However, as will be reviewed later, the results are mixed.

<sup>&</sup>lt;sup>2</sup> Hubbard, Kuttner, Palia (2002) study how banks' health affect borrowers' cost of fund, also using a matched sample of loans. My study differs from theirs in two aspects. First, their sample is restricted to syndicated loans. Second, they do not study the bank effects on firms' investment behavior and stock market valuation.

<sup>&</sup>lt;sup>3</sup> An indirect effect is that, as most of long-term bank loans in Japan are collateralized by land, the credit quality of bank loans thus deteriorated.

<sup>&</sup>lt;sup>4</sup> It is also possible that this result simply reflects the fact that I control for other firm characteristics which may determine firms' access to bond market.

In the NIKKEI NEEDS database, which provides financial data on banks, the land areas owned are also available. Under the assumption that there is a homogenous unit price, it can be converted into the market value. However, the homogenous unit price assumption does not generally hold. For example, a piece of land in the central business district of Tokyo would be much more expensive than a parcel of equal size in the suburbs. The book value of land may better reflect its price and therefore its market value at the time of purchase. Moreover, NIKKEI NEEDS data results in less number of observations. Nevertheless, when I use the land area as an alternative measure of land holding, I obtain similar results.

<sup>&</sup>lt;sup>6</sup> The results are actually stronger, which is not surprising as the GLS is more efficient than the OLS if the model is correctly specified (Hausman tests do not reject the exogeneity of the bank effects). For the sake of brevity, I do not report these results but they are available upon request.

<sup>&</sup>lt;sup>7</sup> Some studies also use the classification of Keiretsu no Kenkyu published by the Keizai Chosa Kyokai (Economic Survey Association). Both Keiretsu no Kenkyu and Dodwell publications

classify Keiretsu firms based loan structure, bank shareholding, and historical factors. Dodwell's definition of group firms is narrower than Keiretsu no Kenkyu and are more stable over time. Using the Dodwell classification, less than 4% of the firms in the sample switch into or out of their groups over a 13-year period. As I already control for whether the bank is the largest lender and other time-varying relationship variables, I prefer a classification that is more stable. In my sample, of all the largest lenders, 23% are main banks.

The loan "ever-greening" explanation of the economic recession in Japan is not inconsistent with the lending channel. In fact, it is very likely that the "ever-greening" incentive at the banks exacerbates the real effects of the bank liquidity shock by reducing the amount of credit available to the manufacturing sector. Indeed, in a Bank of Japan (BOJ) Financial Institution Research Institute Report issued in 2000, the BOJ expressed the concern that, due to increased lending to the real estate sector after the burst of bubbles, banks imposed very tight credit to small and medium enterprises. For more details, see <a href="http://www.imes.boj.or.jp/japanese/kinyu/2000/kk19-4-5.pdf">http://www.imes.boj.or.jp/japanese/kinyu/2000/kk19-4-5.pdf</a> (in Japanese).

<sup>&</sup>lt;sup>9</sup> One way to distinguish between the risk-shifting and the precautionary-saving incentives is to account for the effect of liquidity on banks with different capital positions. As pointed out by Keeley (1990), when banks have less capital at stake, they have little to lose and tend to be more aggressive in extending loans. On the other hand, a weak capital position increases the need for precautionary saving. Therefore, in a later estimation, I include as independent variable an interaction term between liquid assets and a dummy variable indicating whether the firm's capital ratio in 1989 is below the *4%* capital requirement. The results are supportive of the precautionary-saving incentive: the interaction term is significantly negative (columns (3a) and (3b) in Table 2). The coefficient on the liquid assets itself is positive but insignificant.

<sup>10</sup> Basel Accord, an international agreement on bank capital standards, was introduced in 1988.

Although the main focus of the Accord is on risk-based capital requirements, it sets a core (basic) capital requirement of 4%. Given that the data on risk-based capital ratio is not available, I use the core requirement to measure whether the bank faces binding capital requirements.

- <sup>12</sup> The additional control variables in columns (2c) and (2d), namely the interaction between the top-lender dummy and real estate loans and the percent of operating funds are not statistically significant in the first-stage regression. Including them in the first-stage estimation does not change the results.
- <sup>13</sup> One may think that a similar argument can be applied to banks' land holdings. However, banks may have purchased land for speculative purposes or have obtained land through foreclosure. These land holdings are much more recent and use up cash, half of their value evaporated during the collapse.
- <sup>14</sup> Under reasonable assumptions, a detectable real impact generally requires that the bank at least cut back lending proportionally (based on outstanding loans) across lenders.
- <sup>15</sup> For an overview of LAD and quantile regressions in economics research, see Koenker and Hallock (2001). Koenker and Bassett (1978) show that the regression median is more efficient than the OLS estimator in a linear model for any distribution for which the median is more efficient than the mean in the location model.
- <sup>16</sup> Note this sample shrinks from that in the lending equation mainly due to the requirements that (1) the lending relationships existed in 1989; (2) valid firm-level financial information in the DBJ financial database. I am not aware of any systematic pattern in the firms dropped. Indeed, the earlier results in Table 2 can be reproduced in this smaller set of firms (unreported but available upon request), further suggesting that the firm-level results are not driven by sample selection.

<sup>&</sup>lt;sup>11</sup> I thank Murillo Campello for pointing out this possibility to me.

- <sup>20</sup> During 1986-89 when the market performed well, high beta stocks had better performance on average than did low beta stocks. When I regress RET9498 on RET8689, I am effectively regressing the return on the beta. If the beta is constant over time, one would expect high beta stocks to perform poorly, since the market performed poorly during 1994-98.
- This involves calculating the reduction in loan growth due to real estate loans for each bank-firm pair above the 25<sup>th</sup> percentile ( $d_{ij} = 2.138*(\% Real Estate Loans_i 3.6\%)$ , where i indexes banks and j indexes firms) and then aggregating across bank-firm pairs ( $=log(\sum L_{ij}e^{-di}/\sum L_{ij})$ , where L<sub>ij</sub> is the loan balances prior to the shock between bank i and firm j).

<sup>&</sup>lt;sup>17</sup> This is a relatively restrictive cutoff. As it takes time for firms to prepare for bond issuance, they may not be able to issue whenever they get an investment grade. However, alternative cutoffs of meeting the criteria once to five times do not change the results.

<sup>&</sup>lt;sup>18</sup> The significance level of the interaction between land holding and the BIND dummy increases to 5% when I drop this variable, suggesting that this term is adding noise to the estimation. Therefore I drop this variable in the later estimation.

<sup>&</sup>lt;sup>19</sup> I thank the referee for suggesting this term to me.

#### **Table 1. Summary Statistics**

This table presents descriptive statistics on variables used in the paper. Panel A presents summary statistics for relationship variables. Log(loan growth) is the log of the average borrowing from a particular bank during 1994-1998 normalized by the average borrowing from the same bank during 1984-1989. Duration is the number of years that the firmbank pair had positive loan balances in the past 10 years during 1984-1993. Top Lender is whether the bank served as the firm's largest lender at least once during 1984-1993. Equity Stake is the number of shares owned by the bank as a percentage of the firm's total shares outstanding in 1989. Main Bank is a dummy variable indicating whether the bank is the firm's main bank. Panel B presents the statistics for banks. % Real Estate Loans are the loans to the real estate sector pver the total assets in 1989. % Land Holding is the market value of land over the total assets in 1989. q is Tobin's average q. Liquid Assets / Assets is the sum of cash and government security holdings over total assets. Capital Ratio is the book equity over total assets. % Operating Funds is the operating funds as a percentage of total loans in 1989. % Stock Holdings is the equity holdings as a percentage of total assets in 1989. SOLD is a dummy variable indicating whether the bank sold its equity holdings druing years 1989-1993. Panel C presents the statistics for firms. I / K is the average investment rate defined as fixed investments normalized by the beginning-of-period capital stock between 1994 and 1998. q is Tobin's average q. Cash Flow / K is cash flow normalized by normalized by the beginning-of-period capital stock. Cash Stock / K is the cash stock normalized by the beginning-of-period capital stock at the end of 1993. Land/K (firm) is the firm's market value of land normalized by the replacement cost of capital in 1989. % Recent Purchase is the proportion of land (in market value) purchased during 1988-1990. Top Lender's Loan Share is the firm's top lender's share of total long-term

Statistics				Vari	able Names				
Panel A: Ba	nking Relationship V	ariables							
	Log (loan growth)	Duration	Top Lender	Equity Stake (%)	Main bank				
Mean	0.432	6.85	0.174	3.123	0.077				
Median	0.371	7		3.11					
Panel B: Ba	nk Level Variables								
	% Real Estate				Liquid Assets /	Capital Ratio	% Operating Funds		
	Loans	% Land Holding	Log(assets)	Tobin's q	Assets	(1989)	(1989)	% Stock Holdings	SOLD
Mean	0.064	0.079	14.629	1.022	0.278	0.034	0.671	0.022	0.242
Median	0.05	0.048	14.481	1.017	0.258	0.033	0.67	0.019	0.000
Panel C: Fir	m Level Variables								
Variables i	used in the investmen	t regressions:							
						% Recent		Top Lender's Loan	
	I / K	Tobin's q	Cash Flow / K	Cash Stock / K	Land / K (firm)	Purchase	Debt / Asssets	Share (1989)	
Mean	0.306	0.87	0.084	0.187	0.677	0.018	0.171	0.364	
Median	0.086	0.716	0.019	0.165	0.691	0.004	0.164	0.303	
Additional	variables used in the	return regressions	s:						
		Return (1986-	Price Earning	Book to Market		Bank Debt /	flow 1986-89/	Security Holdings /	
	Return (1994-98)	89)	Ratio (1989)	(1989)	Beta	Debt	Assets	Assets	
Mean	-0.262	1.066	49.376	0.236	0.341	0.623	0.021	0.054	
Median	-0.343	0.615	31.29	0.233	0.329	0.702	0.019	0.038	

#### Table 2. Bank Real Estate Exposure and Lending

This table presents the effect of the bank's real estate exposure prior to the land-price collapse on its lending to its client firms based on Heckman two-stage regressions. The first stage is a probit regression. The dependent variable is a dummy variable indicating the survival of the lending relationship. The second stage is an OLS regression. The dependent variable is the log of average borrowing from a particular bank during 1994-1998 normalized by average borrowing from the same bank during the years 1984-1989. *Duration* is the number of years that the firm-bank pair had positive loan balances in the past 10 years during the years 1984-1993. *Top-Lender* is a dummy indicating whether the bank served as the firms largest lender at least once during the years 1984-1993. *Main Bank* is a dummy variable indicating whether the bank is the firm's main bank. % *Equity Stake* is the number of shares owned by the bank as a percentage of the firm's total shares outstanding. q is Tobin's average q. *Liquid Assets / Assets* is the sum of cash and government security holdings over total assets. *Capital Ratio* is the book equity over total assets in 1989. % *Operating Funds* is the operating funds as a percentage of total loans in 1989. % *Real Estate Loans* are the loans to the real estate sector over total assets in 1989. % *Land Holding* is the market value of land over total assets in 1989. *BIND* is a dummy variable indicating whether the top lender's capital ratio in 1989 is below the capital requirement of 4%. % *Stock Holdings* is the equity holdings as a percentage of total assets in 1989. *SOLD* is a dummy variable indicating whether the bank sold its equity holdings druing the years 1989-1993. Heteroscedasticity-consistent standard errors are presented in parentheses. Significance at the 1%, 5%, and 10% levels is indicated by \*\*\*\*, \*\*\*, and \*, respectively.

	(1.)	(11.)	(2.)	(21.)	(2.)	(2.)	(21.)
	(1a) Probit	(1b) OLS	(2a) Probit	(2b) OLS	(2c) OLS	(3a) OLS	(3b) OLS
Relationship Characteristics:	110011	OLD	110011	OLS	OLD	OLS	OLS
Duration	0.300***	-0.137***	0.301***	-0.147***	-0.162***	0.301***	-0.151***
	(0.010)	(0.043)	(0.010)	(0.042)	(0.043)	(0.010)	(0.043)
Top Lender	0.306***	0.090**	0.310***	0.292***	0.255***	0.309***	0.285***
	(0.043)	(0.039)	(0.043)	(0.077)	(0.078)	(0.043)	(0.078)
Main Bank	0.156**	0.355***	0.146**	0.335***	0.346***	0.148**	0.335***
	(0.068)	(0.048)	(0.069)	(0.048)	(0.049)	(0.069)	(0.049)
% Equity Stake	-0.017	-0.015	-0.016	-0.016	-0.015	-0.017	-0.006
	(0.013)	(0.012)	(0.013)	(0.012)	(0.012)	(0.013)	(0.013)
Bank Characteristics:							
Log (assets)	0.024	0.086***	0.026	0.120***	0.095***	0.025	0.094***
m 1 · · ·	(0.027)	(0.028)	(0.030)	(0.030)	(0.033)	-0.03	-0.033
Tobin's q	0.611	0.207	0.352	-0.455	0.204	0.284	0.169
I:: I A / A	(0.517)	(0.638) -0.531***	(0.560)	(0.731) -0.534**	(0.818) -0.983***	-0.558 0.214	-0.873 0.940
Liquid Assets / Assets	0.3 (0.226)	(0.189)	0.29 (0.240)	(0.214)	(0.310)	-0.254	-0.792
Capital Ratio	(0.220)	(0.169)	0.896	10.311**	11.986**	1.325	5.144
Capital Ratio			(4.002)	(4.004)	(5.213)	-4.122	-5.411
% Operating Funds			(4.002)	(4.004)	-0.515**	-4.122	-0.480*
70 Operating I and					(0.246)		-0.247
Liquid Assets / Assets * BIND					(0.240)		-1.451**
Elquid Fissets / Fissets Birth							-0.714
							0.71
Bank Real Estate Exposure in 1989:							
% Real Estate Loans	-2.845***	-2.138***	-2.449***	-2.003**	-1.845**	-2.682***	-3.176**
	(0.633)	(0.741)	(0.687)	(0.784)	(0.799)	(0.816)	(1.618)
% Land Holding	-2.649***	0.91	-4.641**	-0.006	-0.91	-4.515**	-4.189*
	(0.829)	(0.988)	(2.081)	(2.428)	(2.423)	(2.085)	(2.511)
% Real Estate Loans * BIND			-0.939*	-0.318	0.243	-0.897	3.187
			(0.584)	(0.651)	(0.744)	(0.601)	(2.034)
% Land Holding * BIND			2.017	0.994	1.332	1.952	3.918
T I I D *0/ D I E / /	T		(1.753)	(2.248)	(2.299)	(1.750)	(2.411)
Top-Lender Dummy * % Real Estate	Loans			-3.503***	-3.172***		-3.406***
% Stock Holdings				(1.146)	(1.145)	0.931	(1.149) -6.802***
% Stock Holdings						(1.936)	(1.824)
SOLD - Stock Holdings						(1.930)	-0.085
SOLD - Stock Holdings							(0.172)
IMR		1.064***		1.003***	0.897***		0.941***
		(0.284)		(0.282)	(0.284)		(0.284)
Observations	11393		11393	7452		11393	7452
	11393	7452	11393		7452	11393	
Psuedo R <sup>2</sup> / R <sup>2</sup>		0.65		0.65	0.65		0.65

#### Table 3. Effect of the Top Lender's Real Estate Exposure on the Firm's Fixed Investment

This table presents the effect of the top lender's real estate exposure on firm investment based on Least Absolute Distance (LAD) Regressions. The dependent variable I/K is the average investment rate (defined as fixed investments normalized by the beginning-of-period capital stock) between 1994 and 1998. Cash flow / K is the cash flow normalized by normalized by the beginning-of-period capital stock. Cash Stock / K is the cash stock normalized by the beginning-of-period capital stock at the end of 1993. q is Tobin's average q, Land / K (firm) is the firm's market value of land normalized by the replacement cost of capital in 1989. % Recent Purchase is the proportion of land (in market value) purchased during 1988-1990. Leverage is the debt to assets ratio. High Leverage is a dummy variable indicating a firm's leverage above the median. % Real Estate Loans is the loans to the real estate sector normalized by the total assets in 1989. % Land Holding is market value of land normalized by the total assets in 1989 for the top lender. CON is a dummy variable indicating whether the top lender's loan share is above the median. BIND is a dummy variable indicating whether the top lender's capital ratio in 1989 is below the capital requirement 4%. Standard errors are calculated based on the asymptotic variance. They are presented in parentheses. Significance at the 1%, 5%, and 10% levels is indicated by \*\*\*, \*\*\*, and \*, respectively.

q         0.025***         0.025***         0.025***         0.025***         0.026***           Cash Flow / K         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.001)         (0.022)         -0.016         (0.022)         -0.016         (0.022)         -0.016         (0.003)         (0.004)         (0	, <u>r</u>	(1)	(2)	(3)	(4)	(5)
Cash Flow / K         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.000)         (0.002)         (0.001)         (0.002)         (0.001)         (0.002)         (0.001)         (0.002)         (0.001)         (0.002)         (0.004)         (0.003)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.006)         (0.006)         (0.006)         (0.006)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)         (0.004)	a					
Cash Flow / K         1.111***         1.116***         1.107***         1.083***         1.083***         1.083***         0.0020         0.0101         0.0202         0.0101         0.0202         0.0101         0.0202         0.0104         0.0206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.206***         0.0002         0.0003         0.0003         0.0003         0.0003         0.0003         0.0003         0.0003         0.0003         0.0003         0.0004         0.0003         0.0004         0.0014         0.0014         0.0014         0.0014         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0024         0.0020         0.0028         0.0048         0.0049         0.0048         0.0049         0.0048         0.0022         0.0023         0.0023         0.0016**         0.0024         0.0023         0.0023         0.0023         0.0023         0.0023         0.0023         0.0023         0.0023         0.0023         0.0023	Ч					
Cash Stock / K	Cash Flow / K					
Cash Stock / K         0.205***         0.204***         0.205***         0.206***         0.206***           (0.003)         (0.003)         (0.003)         (0.003)         (0.003)         0.003         -0.004           Land / K 1989 (firm)         -0.132***         -0.138***         -0.173***         -0.168***           (0.036)         (0.044)         (0.043)         (0.044)         -0.031           % Recent Purchase         -0.064         -0.07         -0.042         -0.027         -0.024           Leverage         0.090*         0.116*         0.110**         0.157**         0.158***           Leverage         0.090**         0.116*         0.140**         0.157**         0.118***           Land / K 1989 (firm) * High Leverage         0.092***         -0.097***         -0.118***         -0.107***         -0.117***         -0.117***         -0.117***         -0.117***         -0.117***         -0.117***         -0.117***         -0.117***         -0.010**         -0.020**         (0.023)         (0.023)         (0.013)         -0.020**         -0.012***         -0.012***         -0.012***         -0.024**         -0.012***         -0.12***         -0.12***         -0.02**         -0.12***         -0.02**         -0.02**         -0.02**						
Control   Cont	Cash Stock / K		` /	` /	,	
Land / K 1989 (firm)         -0.132***         -0.138***         -0.173***         -0.173***         -0.168***           Recent Purchase         -0.064         -0.07         -0.042         -0.027         -0.024           Recent Purchase         -0.064         -0.07         -0.042         -0.027         -0.024           Leverage         0.090*         0.116*         0.140**         0.157***         0.154***           Land / K <sup>1980</sup> (firm) * High Leverage         -0.092***         -0.097**         -0.118***         -0.118***         -0.117***         -0.117***           Access to the Public Bond Market         -0.018         (0.022)         (0.022)         -0.023         (0.016)           Seal Estate Exposure and Other Characteristics in 1982*         -0.014**         (0.004)         -0.013         -0.017***           Real Estate Loans         -0.615***         -0.509*         -0.664**         -0.640*         -0.712***           Real Estate Exposure and Other Characteristics in 1982*         -0.013**         -0.030*         0.0309         0.0369         0.0259           Real Estate Loans         -0.016**         -0.509*         -0.664**         -0.640*         -0.712***           Real Estate Loans Sanc         -0.229*         2.218         2.21         2.21<			(0.003)	(0.003)		-0.002
Recent Purchase         (0.036)         (0.044)         (0.043)         (0.044)         -0.07           We Recent Purchase         -0.064         -0.07         -0.042         -0.027         -0.024           Leverage         (0.090)         0.116*         (0.140)*         0.157***         0.154***           Leverage         (0.094)         (0.067)         (0.069)         (0.068)         (0.049)           Land / K <sup>1989</sup> (firm) * High Leverage         -0.092***         -0.097***         -0.118***         -0.107***         -0.117***           Access to the Public Bond Market         -0.018         (0.022)         (0.022)         (0.023)         (0.018)           Access to the Public Bond Market         -0.618**         -0.014         (0.004)         -0.072***         -0.118***         -0.107***         -0.118***         -0.023         (0.018)         0.022         (0.023)         0.023         0.011***         -0.011***         -0.023         0.011***         -0.023         0.018**         -0.023         0.023         0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024         -0.024	Land / K <sup>1989</sup> (firm)	· · · ·	, , ,	, ,		-0.168***
% Recent Purchase         -0.064         -0.07         -0.042         -0.027         -0.024           Leverage         (0.118)         (0.145)         (0.111)         (0.109)         -0.08           Leverage         (0.054)         (0.067)         (0.069)         (0.068)         (0.049)           Land / K <sup>1989</sup> (firm) * High Leverage         -0.092***         -0.097***         -0.118***         -0.107***         -0.117***           Access to the Public Bond Market         -0.001**         (0.022)         (0.023)         (0.016)           Access to the Public Bond Market         -0.615**         -0.509*         -0.664**         -0.640*         -0.712***           % Real Estate Exposure and Other Characteristics in 1985:         (0.249)         (0.305)         (0.309)         (0.369)         (0.265)           % Real Estate Loans         -0.615**         -0.509*         -0.664**         -0.640*         -0.712***           % Land Holding         -0.07         -0.134         -0.484         -0.001         -0.099           % Lander's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Top Lender's Capital Ratio         2.520*         2.18         2.11         -0.104**         -0.114***	Zuita / 12 (11111)					
Content	% Recent Purchase					
Leverage         0.090* (0.054)         0.116* (0.067)         0.140** (0.068)         0.154*** (0.049)           Land / K <sup>1980</sup> (firm) * High Leverage         -0.092*** (0.022)         -0.118*** (0.022)         -0.107*** (0.023)         -0.117***           Access to the Public Bond Market         -0.018 (0.022)         (0.022)         (0.023)         (0.014)           Experiment in Public Bond Market         -0.014 (0.004)         (0.014)         (0.004)         -0.014           Experiment in Public Bond Market         -0.615**         -0.509*         -0.664**         -0.60**         -0.712***           Real Estate Exposure and Other Characteristics in 1989:         -0.664**         -0.640*         -0.712***           Real Estate Loans         -0.615**         -0.509*         -0.664**         -0.60**         -0.712***           Real Estate Loans         (0.249)         (0.305)         (0.309)         (0.369)         (0.265)           Land Holding         -0.07         -0.134         -0.484         -0.014         (0.072)         -0.112***           Top Lender's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Real Estate Loans * Share         -0.169**         -0.123**         -0.109**         -0.114***           Real Estate						
Control   Cont	Leverage					
Count   Coun		(0.054)	(0.067)	(0.069)	(0.068)	(0.049)
Count   Coun	Land / K <sup>1989</sup> (firm) * High Leverage	-0.092***	-0.097***	-0.118***	-0.107***	-0.117***
Access to the Public Bond Market         -0.014 (0.014)         (0.004)         (0.0013)           Top Lender's Real Estate Exposure and Other Characteristics in 1989:           % Real Estate Loans         -0.615**         -0.509*         -0.664**         -0.640*         -0.712***           % Real Estate Loans         -0.015**         -0.509*         -0.664**         -0.640*         -0.712***           % Land Holding         -0.07         -0.134         -0.484         0.001         -0.009           Lender's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Top Lender's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Top Lender's Loan Share         (1.419)         (1.694)         (1.738)         (2.343)         (1.379)           % Real Estate Loans * CON         0.075         0.075         0.075         0.075         0.075           % Land Holding * CON         0.487         0.075         0.075         0.075         0.075           % Land Holding * BIND         0.487         0.075         0.075         0.075         0.075           % Land Holdings         0.075         0.075         0.075         0.075         0.075	, , , , , , , , , , , , , , , , , , , ,					
Top Lender's Real Estate Exposure and Other Characteristics in 1989:           % Real Estate Loans         -0.615**         -0.509*         -0.664**         -0.640*         -0.712***           % Real Estate Loans         -0.615**         -0.509*         -0.664**         -0.640*         -0.712***           % Land Holding         -0.07         -0.134         -0.484         0.001         -0.009           % Land Holding         -0.0292         (0.352)         (0.482)         -0.514         (0.372)           Top Lender's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Top Lender's Loan Share         -0.123***         -0.102***         -0.114***           % Real Estate Loans * CON         0.075         -0.075           % Land Holding * CON         0.487         -0.044           % Real Estate Loans * BIND         -0.044         -0.044           % Real Estate Loans * BIND         -0.044         -0.044           % Land Holding * BIND         -0.044         -0.075           % Stock Holdings         -0.075         -0.075           % Stock Holdings         -0.075         -0.075           Column Colu	Access to the Public Bond Market	,		, ,	,	, ,
% Real Estate Loans       -0.615**       -0.509*       -0.664**       -0.640*       -0.712***         (0.249)       (0.305)       (0.309)       (0.369)       (0.265)         % Land Holding       -0.07       -0.134       -0.484       0.001       -0.009         (0.292)       (0.352)       (0.482)       -0.514       (0.372)         Top Lender's Capital Ratio       2.520*       2.218       2.811       5.654**       5.320***         Top Lender's Loan Share       (1.419)       (1.694)       (1.738)       (2.343)       (1.379)         % Real Estate Loans * CON       (0.037)       -0.102***       -0.114***         % Land Holding * CON       0.075       0.075       -0.029       (0.021)         % Real Estate Loans * BIND       0.487       0.705       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.075       -0.044       -0.075       -0.044       -0.075 <td></td> <td></td> <td></td> <td>, ,</td> <td></td> <td></td>				, ,		
% Real Estate Loans       -0.615**       -0.509*       -0.664**       -0.640*       -0.712***         (0.249)       (0.305)       (0.309)       (0.369)       (0.265)         % Land Holding       -0.07       -0.134       -0.484       0.001       -0.009         (0.292)       (0.352)       (0.482)       -0.514       (0.372)         Top Lender's Capital Ratio       2.520*       2.218       2.811       5.654**       5.320***         Top Lender's Loan Share       -0.123***       -0.103***       -0.100***       -0.114***         % Real Estate Loans * CON       0.075       0.075       -0.029       (0.021)         % Land Holding * CON       0.487       (0.705)       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.044       -0.055       -0.044       -0.075       -0.044       -0.075       -0.044       -0.075       -0.044       -0.075       -0.044       -0.075       -0.075       -0.075       -0.075       -0.075       -0.075       -0.075       -0.075       -0.075       -0.075       -0.						
% Land Holding       (0.249)       (0.305)       (0.309)       (0.369)       (0.265)         % Land Holding       -0.07       -0.134       -0.484       0.001       -0.009         (0.292)       (0.352)       (0.482)       -0.514       (0.372)         Top Lender's Capital Ratio       2.520*       2.218       2.811       5.654**       5.320***         Top Lender's Loan Share       -0.123***       -0.100***       -0.114***         % Real Estate Loans * CON       0.075       0.075       0.029       (0.021)         % Land Holding * CON       0.487       0.487       0.075       0.044       0.048       0.057       0.057       0.057       0.057       0.057       0.057       0.057       0.057       0.070       0.070       0.070       0.070       0.070       0.070       0.0	Top Lender's Real Estate Exposure and Other	er Characteristics in 19	<u>89:</u>			
% Land Holding       -0.07       -0.134       -0.484       0.001       -0.009         Top Lender's Capital Ratio       2.520*       2.218       2.811       5.654**       5.320***         Top Lender's Capital Ratio       2.520*       2.218       2.811       5.654**       5.320***         Top Lender's Loan Share       -0.123***       -0.100***       -0.114***         (0.037)       -0.029       (0.021)         % Real Estate Loans * CON       0.075       -0.075         % Land Holding * CON       0.487       -0.044         % Real Estate Loans * BIND       -0.044       -0.044         % Land Holding * BIND       -0.044       -0.075         % Stock Holdings       -0.075       -0.075         % Stock Holdings       -0.075       -0.075         COLD - Stock Holdings       -0.075       -0.075         Observations       420       420       420       420       420	% Real Estate Loans	-0.615**	-0.509*	-0.664**	-0.640*	-0.712***
COUNTION   COUNTION		(0.249)	(0.305)	, ,	(0.369)	` ′
Top Lender's Capital Ratio         2.520*         2.218         2.811         5.654**         5.320***           Top Lender's Loan Share         (1.419)         (1.694)         (1.738)         (2.343)         (1.379)           Top Lender's Loan Share         -0.123***         -0.100***         -0.114***           (0.037)         -0.029         (0.021)           % Real Estate Loans * CON         (0.132)         (0.132)           % Land Holding * CON         0.487         (0.705)           % Real Estate Loans * BIND         -0.044         (0.145)           % Land Holding * BIND         -1.118**         -1.189*         -1.113**           % Stock Holdings         -0.075         -0.075         -0.075           SOLD - Stock Holdings         -0.075         -0.075         -0.075           Observations         420         420         420         420         420         420	% Land Holding					
Company					-0.514	
Top Lender's Loan Share       -0.123***       -0.100***       -0.114***         (0.037)       -0.029       (0.021)         % Real Estate Loans * CON       0.075       (0.132)         % Land Holding * CON       0.487       -0.044         % Real Estate Loans * BIND       -0.044       -0.044         % Land Holding * BIND       -1.189*       -1.113**         % Stock Holdings       -0.075       -0.075         % Stock Holdings       -0.070       -0.070         SOLD - Stock Holdings       -0.070       -0.070         Observations       420       420       420       420       420	Top Lender's Capital Ratio				5.654**	5.320***
(0.037)		(1.419)	(1.694)			
% Real Estate Loans * CON       0.075         % Land Holding * CON       0.487         (0.705)       (0.705)         % Real Estate Loans * BIND       -0.044         % Land Holding * BIND       -1.189*       -1.113**         % Stock Holdings       (0.693)       (0.557)         % SOLD - Stock Holdings       0.075         SOLD - Stock Holdings       0.075         Observations       420	Top Lender's Loan Share					
Marcology   Marc					-0.029	(0.021)
% Land Holding * CON       0.487 (0.705)         % Real Estate Loans * BIND       -0.044 (0.145)         % Land Holding * BIND       -1.189* -1.113**         % Stock Holdings       -0.075         % Stock Holdings       -0.075         SOLD - Stock Holdings       -0.075         Observations       420	% Real Estate Loans * CON					
(0.705)  % Real Estate Loans * BIND  % Land Holding * BIND  % Land Holdings  % Stock Holdings  SOLD - Stock Holdings  Observations  (0.705)  (0.702)  (0.693)  (0.557)  (0.702)  (0.702)  (0.702)  (0.702)  (0.062)	0/ I - 111 11					
% Real Estate Loans * BIND       -0.044         % Land Holding * BIND       -1.189*       -1.113**         % Stock Holdings       (0.693)       (0.557)         SOLD - Stock Holdings       (0.702)         SOLD - Stock Holdings       0.07         Observations       420	% Land Holding * CON					
(0.145)   (0.145)   (0.147)   (0.148)   (0.148)   (0.693)   (0.557)   (0.693)   (0.557)   (0.693)   (0.702)   (0.702)   (0.702)   (0.702)   (0.702)   (0.702)   (0.693)   (0.570)   (0.702)   (0.702)   (0.693)   (0.702)   (0.693)   (0.702)   (0.693)   (0.570)   (0.702)   (0.693)   (0.570)   (0.693)   (0.570)   (0.693)   (0.570)   (0.693)   (0.570)   (0.693)   (0.6	0/ Deal Estate Leans * DIND			(0.705)	0.044	
% Land Holding * BIND       -1.189*       -1.113**         % Stock Holdings       (0.693)       (0.557)         % Stock Holdings       (0.702)         SOLD - Stock Holdings       (0.702)         Observations       420	% Real Estate Loans " BIND					
% Stock Holdings         (0.693)         (0.557)           SOLD - Stock Holdings         0.075           SOLD - Stock Holdings         0.070           Observations         420 <td>% I and Holding * RIND</td> <td></td> <td></td> <td></td> <td></td> <td>1 112**</td>	% I and Holding * RIND					1 112**
% Stock Holdings     0.075       SOLD - Stock Holdings     0.0702       SObservations     420     420     420     420     420     420     420     420	70 Land Holding * BIND					
(0.702)   SOLD - Stock Holdings   (0.702)   Observations   420	% Stock Holdings				(0.073)	
SOLD - Stock Holdings     0.07       Compared to the comparison of the c	70 Stock Holdings					
Observations 420 420 420 420 420 420 420	SOLD - Stock Holdings					
Observations 420 420 420 420 420 420	2022 Stock Holdings					
Psuedo R <sup>2</sup> 0.44 0.45 0.45 0.45	Observations	420	420	420	420	
	Psuedo R <sup>2</sup>	0.44	0.44	0.45	0.45	0.45

#### Table 4. The Top Lender's Real Estate Exposure and Firm Valuation

This table presents the effect of the top lender's real estate exposure on firm valuation based on OLS regressions. The dependent variable is the buy-and-hold stock return between 1994 and 1998. *Return* (1986-1989) is the buy-and-hold stock return between 1986 and 1989. *Beta* is calculated using monthly return during 1986-1989. *Cash Flow* (1986-89) / Assets is the cumulative cash flow between 1986 and 1989 normalized by the total assets in 1989. *Land/K* (firm) is the firm's market value of land normalized by the replacement cost of capital in 1989. \*\*Real Estate Loans\*\* are the loans to the real estate sector normalized by the total assets in 1989. \*\*Land Holding\*\* is market value of land normalized by the total assets in 1989 for the top lender. *CON* is a dummy variable indicating whether the top lender's loan share is above the median. *BIND* is a dummy variable indicating whether the top lender's capital ratio in 1989 is below the capital requirement of 4%. Heteroscedasticity-consistent standard errors are presented in parentheses. Significance at the 1%, 5%, and 10% levels by a two-sided test is indicated by \*\*\*, \*\*\*, and \*, respectively. Significance at the 10% and 15% level by a one-sided test is indicated by a and b respectively.

	(1)	(2)	(3)	(4)	(5)
Return (1986-89)	-0.039***	-0.039***	-0.039***	-0.041***	-0.041***
_	(0.011)	(0.012)	(0.012)	(0.011)	-0.011
Beta	0.046	0.048	0.041	0.042	0.042
<b>2.1.71</b> (400.400)	(0.036)	(0.037)	(0.037)	(0.037)	-0.037
Cash Flow (1986-89) / Assets	0.364	1.495*	1.606**	1.553*	1.554*
	(0.497)	(0.789)	(0.816)	(0.831)	-0.832
Leverage (1993)	0.441***	0.404**	0.361*	0.399**	0.400**
	(0.170)	(0.185)	(0.185)	(0.178)	-0.178
Bank Debt / Total Debt (1993)	0.085*	0.065	0.058	0.076	0.076
	(0.050)	(0.053)	(0.054)	(0.053)	-0.053
Price Earnings Ratio (1989)	0.000	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Book to Market (1989)	0.007***	0.005*	0.006*	0.006**	0.006**
	(0.003)	(0.003)	(0.003)	(0.003)	-0.003
Land / K <sup>1989</sup> (firm)	-0.266**	-0.315***	-0.298**	-0.290**	-0.289**
	(0.114)	(0.121)	(0.124)	(0.124)	-0.126
% Recent Purchase	0.124	0.106	0.127	0.143	0.142
	(0.182)	(0.184)	(0.186)	(0.185)	-0.185
Security Holdings / Assets (1993)		-0.002	0.027	0.051	0.052
		-0.268	(0.271)	(0.274)	-0.278
Export Sales / Sales (1993)		0.000	0.000	0.000	0.000
•		(0.000)	(0.000)	(0.000)	(0.000)
Access to the Public Bond Market		-0.035	-0.031		
		(0.037)	(0.037)		
Top lender's Real Estate Exposure and Other	Characteristics in 1989	9:			
% Real Estate Loans	0.307	0.377	0.079	0.809	0.846
70 Real Estate Boards	(0.750)	-0.764	(0.856)	-0.964	-1.105
% Land Holding	-1.516**	-1.502**	-1.917***	-1.426*	-1.679**
70 Eand Holding	(0.697)	(0.710)	(0.721)	(0.786)	(0.853)
Top Lender's Capital Ratio	-3.8	-2.706	-3.711	3.64	3.439
Top Bender's Cupital Tanks	(4.134)	(4.163)	(4.276)	(5.432)	(5.527)
Top Lender's Loan Share	()	(	-0.186*	-0.130*	-0.131*
r			(0.098)	(0.077)	(0.077)
% Real Estate Loans * CON			0.608	` '	, ,
			(0.770)		
% Land Holding * CON			0.797		
-			(1.702)		
% Real Estate Loans * BIND				$-1.190^{a}$	-1.246 <sup>a</sup>
				(0.909)	(0.875)
% Land Holding * BIND				-0.43	, ,
2				(1.289)	
Top Lender's Stock Market Exposure					0.030
· ·					(2.460)
SOLD Equity Holdings					0.008
					(0.243)
Observations	476	448	448	448	448
$R^2$	0.12	0.13	0.13	0.13	0.13